Author’s response to reviews

Title: The differences in the consumption of proteins, fats and carbohydrates in the diet of pregnant women diagnosed with arterial hypertension or arterial hypertension and hypothyroidism

Authors:

Wioletta Waksmańska (wwaksmanska@ath.bielsko.pl)
Rafał Bobiński (rbobinski@ath.bielsko.pl)
Iabela Ułman-Włodarz (i.ulmanwiodarz@gmail.com)
Anna Pielesz (apielesz@ath.bielsko.pl)

Version: 2 Date: 28 Apr 2019

Author’s response to reviews:

1. The abstract was improved.

2. The heading "Declarations" was added as well as other specific sub-sections:

   - Ethics approval and consent to participate
   - Consent for publication
   - Availability of data and material
   - Competing interests
   - Funding
   - Authors’ contributions
   - Acknowledgements.

3. The Ethics approval and consent to participate section were improved.

4. The full name of the ethics committee was included in the Ethics approval and consent to participate section.

All the changes and adjustments were made according to information included in the email sent by BMC Pregnancy and Childbirth and in compliance with the submission guidelines.